Author's response to reviews

Title: Healthcare professionals' intentions to use wiki-based reminders to promote best practices in trauma care: A survey protocol

Authors:

Patrick M Archambault (patrick.m.archambault@gmail.com)
France Légaré (france.legare@mfa.ulaval.ca)
André Lavoie (andre.lavoie.trauma.cha@ssss.gouv.qc.ca)
Marie-Pierre Gagnon (marie-pierre.gagnon@fsi.ulaval.ca)
Jean Lapointe (jcoutjla@videotron.ca)
Sylvie St-Jacques (sylvie.st-jacques@crsfa.ulaval.ca)
Julien Poitras (julienpoitras@videotron.ca)
Karine Aubin (Karine_Aubin@ssss.gouv.qc.ca)
Sylvain Croteau (sylvain.croteau@gmail.com)
Martin Pham-Dinh (martinphamdinh@gmail.com)

Version: 3 Date: 30 May 2010

Author's response to reviews: see over
Sunday, May 30, 2010

Dear Drs Eccles and Mittman,

I am submitting a revised version of the study protocol intitled: *Healthcare professionals’ intentions to use wiki-based reminders to promote best practices in trauma care: A survey protocol.*

In response to the Editorial Team’s request, I have made the following changes:

**Major revisions:**

1) All authors names are written exactly as they are in the submission system.
2) The indication that ‘all authors have read and approved the final manuscript’ is now specified in the authors contribution section.

**Minor revisions:**

1) All authors names are abbreviated in the competing interest section.
2) ‘Table 1’ is now intitled ‘Appendix 1’ and is found after ‘authors’ contribution’.
3) I proof read the manuscript and the figures 1 and 2. I have made a few corrections.
   a. In the ‘Participants’ section, I have removed the mention of ‘primary’, ‘secondary’ and ‘tertiary’ trauma centers and replaced these terms respectively by ‘level III’, ‘level II’ and ‘level I’ trauma centers to eliminate any confusion that these two different ways to categorize trauma centers could have.
   b. In the ‘Phase two: Developing the questionnaire’ section, I added an ‘s’ to ‘attitudes’ and ‘subjective norms’.
   c. In the ‘Characteristics of healthcare professionals’ section, I removed the terms ‘primary’, ‘secondary’ and ‘tertiary’ and replaced them respectively by the terms ‘level III’, ‘level II’ and ‘level I’.
4) In Figure 1, I added an ‘s’ to ‘attitudes’ and ‘subjective norms’. I also increased the size of the boxes around the text since some of the text was being cut off.
5) In Figure 2, I increased the size of certain boxes as well because some of the text was being cut off by the boxes.

Sincerely,

Patrick Archambault, M.D., MSc., FRCPC
Corresponding author
Emergency and Critical Care Medicine Specialist
Centre Hospitalier Affilié Universitaire Hôtel-Dieu de Lévis
Post-doctoral Fellow
Research center of the Centre hospitalier universitaire de Québec